Author's response to reviews

Title: Bipolar radiofrequency ablation is useful for treating atrial fibrillation combined with heart valve diseases

Authors:

Lin Chen Dr (chenlin853@yahoo.cn)
Yingbin Xiao Dr (xiaoyb@vip.sina.com)
Ruiyan Ma Dr (maruiyan2008@hotmail.com)
Baicheng Chen Dr (chen baikcheng@163.net)
Jia Hao Dr (haojia1973@sina.com)
Chuan Qin Dr (qinchuan1977@hotmail.com)
Wei Cheng Dr (chengwei1974@sina.com)
Renguo Wen Dr (Rengguo1974@hotmail.com)

Version: 2 Date: 14 March 2013

Author's response to reviews: see over
Dear editor,

We have changed the manuscript as requested.

(1) We have added the contributions of every author. Please see the Title Page of the manuscript.

(2) We have added sentences "Prior written and informed consent were obtained from every patient and the study was approved by the ethics review board of Xinqiao Hospital, Third Military Medical University." in the Materials and Methods section.

(3) We have added “All authors declare no financial competing interests. All authors declare no non-financial competing interests.” in page 12.

Sincerely,

Lin Chen
Department of Cardiovascular Surgery, 7F, Second in-Patient Building,
183 Xinqiao St, Shapingba District, Chongqing, 400037, P. R. China.
Tel: 86-23-68774607
Fax: 86-23-68774607
Email: chenlin853@yahoo.cn